Author’s response to reviews

Title: Accuracy of self-reported body weight, height and waist circumference in a Dutch overweight working population

Authors:

Johanna C Dekkers (c.dekkers@vumc.nl)
Marieke F van Wier (m.vanwier@vumc.nl)
Ingrid JM Hendriksen (ingrid.hendriksen@tno.nl)
Jos WR Twisk (jwr.twisk@vumc.nl)
Willem van Mechelen (w.vanmechelen@vumc.nl)

Version: 7 Date: 16 October 2008

Author’s response to reviews: see over
Re: Resubmission of the manuscript 'Accuracy of self-reported body weight, height and waist circumference in a Dutch overweight working population'

Dear editor,

Please find enclosed a revised version of the manuscript "Accuracy of self-reported body weight, height and waist circumference in a Dutch overweight working population" by J Caroline Dekkers et al.

As requested, we made the following formatting changes:

- We made sure that the Abstract present in the manuscript matches exactly the one that is in the submission system by copying the Abstract in the manuscript and pasting it into the submission system.

- We don’t quite understand the formatting changes requested on the author’s contributions: the author’s contributions are on page 16 of the manuscript.

- The submitted Tables were already in portrait format, but now we have deleted visible vertical lines from all Tables. Some of the horizontal lines were deleted as well.

- We changed Table 6a into Table 6 and Table 6b into Table 7 and updated the text accordingly.

- We deleted all information that should not be included in the figure, like Figure titles and numbers.

We hope that the requested formatting changes we made meet your expectations. We are looking forward to the publication of our manuscript in your journal.

Yours sincerely,
Caroline Dekkers, PhD
EMGO Institute and Department of Public and Occupational Health
VU University medical center,
Van der Boechorststraat 7,
1081 BT Amsterdam, The Netherlands
Phone: +31 (20) 444 2806
Fax: +31 (20) 444 8387
e-mail: c.dekkers@vumc.nl